

Author's response to reviews

Title:Analysis and reporting of stepped-wedge randomised-controlled trials: synthesis and critical appraisal of published studies, 2010-2014

Authors:

Calum Davey (calum.davey@lshtm.ac.uk)
James Hargreaves (James.Hargreaves@lshtm.ac.uk)
Jennifer A Thompson (Jennifer.Thompson@lshtm.ac.uk)
Andrew J Copas (a.copas@ucl.ac.uk)
Emma Beard (e.beard@ucl.ac.uk)
James J Lewis (James.Lewis@lshtm.ac.uk)
Katherine L Fielding (Katherine.Fielding@lshtm.ac.uk)

Version:2**Date:**2 June 2015

Author's response to reviews:

Friday, 29 May 2015

Dear Professor Torgerson

Thank you for admitting our paper for peer review, and for giving us additional time to address comments. Please pass on our thanks to the peer reviewers for their insightful comments.

I have addressed their comments in the tables overleaf, and referenced the changes that we made in the re-submission of the manuscript.

Best wishes,

Calum Davey